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Weekly Status Week of July 3 – July 9, 2008
Day of Year 185 - 191
Mission Day 2789 - 2795
Earth Observing-One (EO-1) General

There were 112 Data Collection Events (DCEs) scheduled this week.  
INSTRUMENTS

All instruments operated nominally this week.
Instrument decontamination cycle:
· ALI out-gassing was performed on July 5 (187) from 05:35z to 20:25z.
· Hyperion de-icing was performed on July 5 (187) from 05:25z to 20:15z.

ALI internal calibration (Type II) was performed on July 7 (189) at 00:09z.
EO-1 Spacecraft Subsystems

Command and Data Handling (C&DH)

Continued to experience problems playing back engineering data from its Solid-State-Recorders.

Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations
The weekly teleconference between the Advanced Information System Technology (AIST) Sensor Web collaborators was held on Thursday, July 3.  The following items of interest were discussed:
· Long, long discussion about errors in ALI Level 1R images and what can and cannot be done to correct them. ??????
· There are issues with pixel to pixel co-registration and band to band co-registration erors. ???

· There is concern that only good images be sent to users because they rely on information gleaned from the images to issue alerts and allocate resources to combat threats and distribute aid to victims.
· For example, the Forest Service requires that wildfire predicts be certified.  That raises the issue of how EO-1 products can be certified.
· It was stated that the three EO-1 products useful for viewing wildfires and their resulting effect are the visible bands for viewing (Infrared ?), ALI (?) bands 9, 6 & 4 (SWIR) for viewing burn scars, and the NDVI (Normalized Difference Vegetation Index) and NDBI (Normalized Difference Build-up Index) (?) to view vegetation health.
· The question was raised as to which technique is best to mitigate the ALI image errors. ??????
· There was an action given to summarize the discussed ALI image errors.

· An action was given to call Jeff Mendenhall at Lincoln Lab at MIT to get his opinion on the discussed ALI image errors.

· An action was given to determine who should run the ALI Level 1R for our team.

· It was concluded that we need to evaluate whether we should use Level 1R or Level 1G for EO-1 products.
· Steve Ungar emphasized that for science use bad pixels should not be “fixed” because such action can result in “bad” science results.
· Steve Ungar emphasized that push-broom sequentially sampled instruments cannot ?????
· An action was given to study a USGS ALI Level 1G scene to see if it contains the same error as detected in our Level 1G scenes.  
· Stu Frye took an action to investigate the overall issue of ALI image errors.

Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week with the exception of the period when the WS1 antenna testing was performed.
EO-1 MISSION OPERATIONS CENTER

Real-Time

Spacecraft state of health is currently nominal.
Mission Planning
Mission planning functions are being performed using older versions of MOPSS on Bullseye system.

ASPEN upgrade: 

· Tracking tool generated for procurement of software and hardware

· Awaiting necessary software for upgrades to continue

Flight Dynamics

Orbit prediction and maneuver planning parallel operations:
· Primary operations has begun on new flight dynamics system for orbit prediction 

System Administration

Management System internal audit

· Prepared for audit to ensure compliance with GSFC Management System

GROUND AND SPACE NETWORK

Station Downtimes
Alaska Ground Station (AGS) EOS Data and Operations System (EDOS) fiber relocation:
· On July 10, the EDOS fiber at AGS will be relocated to PF1/PF2

· Supports at AGS are no longer being scheduled on nominal basis

· AGS will be available until scheduled date of July 30 to support critical activities if necessary.
Operational Discrepancies

No major problems to report.

Support Enhancements  
Certification of PF2:
· FOT requested and successfully performed X-band test support with PF2 on July 2 (184) at 19:46z

· X-band test support on July 7 (189) at 19:41z was impacted by NISN line outage

· MOC did not receive telemetry until four minutes after AOS

· Support successful otherwise

· X-band test support conducted on July 8 (190) at 18:42z

· Awaiting verification of X-band data

WS1 antenna testing:
· EO-1 FOT has successfully performed two two-way Doppler tests for the LRO project on July 2 (184) at 16:37z and on July 3 (185) at 17:15z
· No commanding performed during test - simply turned on EO-1 transponder

Upcoming Events
Lunar calibration

· A nominal (all instrument) lunar calibration will be performed on July 18, 2008 during the 23:28z umbra

· A modified lunar calibration will be conducted two orbits later.
MLT control inclination maneuvers.

· First inclination maneuver burn on Aug. 12, 2008.

· Second inclination maneuver burn on Aug. 14, 2008.   

· The MLT should become essentially 10:00 am after these burns.
Imagery Status
Scenes and Engineering Cals planned for week of July 3 – July 9, 2008              112
Total scenes and engineering calibrations planned for entire mission – approximately 41,120
Total Scenes:  ALI scenes in the Level 0 archive            36,XXX (as of July 9, 2008)

                        Hyperion scenes in the Level 0 archive    36,XXX
Publications and Presentations Status (as of 12-05-07)
404 publications 

284 external presentations

53 articles and press releases
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